Author's response to reviews

Title: EXIT procedure in twin pregnancy: a series of three cases from a single center.

Authors:

Lutgardo L García-Díaz (lutgardogd@hotmail.com)
Juan Carlos JC De Agustín (juanc.agustin.sspa@juntadeandalucia.es)
Antonio A Ontanilla (antonio.ontanilla.sspa@juntadeandalucia.es)
Maria Luisa ML Marenco (marial.marenco.sspa@juntadeandalucia.es)
Antonio A Pavon (apavond@telefonica.net)
Antonio A Losada (antonio.losada.sspa@juntadeandalucia.es)
Guillemo A Antiñolo (guillermo.antinolo.sspa@juntadeandalucia.es)

Version: 2
Date: 3 June 2014

Author's response to reviews: see over
June 3\textsuperscript{rd}, 2014

Re: (Manuscript ID 1516169111268048): "EXIT procedure in twin pregnancy: a series of three cases from a single center.

Dear Editors

Thank you for your mail of May 23\textsuperscript{rd} containing indicating us some important items to be completed before taking into consideration our paper for review. We have amended all those points, highlighted in red along the text, as follows:

- **Requesting ethics statement**
  We have included the following paragraph within the Patients and Methods section: “An informed consent was obtained from all the patients for clinical studies. The study and procedures conformed to the tenets of the declaration of Helsinki and were approved by the Institutional Review Boards from the University Hospital Virgen del Rocío of Seville.”

- **Competing Interest**
  A declaration of no competing interests has been included in the corresponding section, following the Discussion section as indicated.

- **Authors’ contribution**
  We have included a section with the contribution of each author, after the Competing interest section

- **Acknowledgements**
  This section has not been included in the ms because everyone who has contributed to the work here presented is included in the authorship list.

- **Requesting for line numbering and for E-mail-add**
  Both amendments have been performed

We hope you will find the changes appropriate and that our manuscript may now be suitable for consideration in BMC Pregnancy and Childbirth.

Thank you very much for your time and consideration.

Yours sincerely,

Guillermo Antiñolo, MD, PhD
Unidad de Gestión Clínica de Genética, Reproducción y Medicina Fetal
Hospitales Universitarios Virgen del Rocío,
Avenida Manuel Siurot s/n, 41013 Sevilla, Spain.
Voice:+34 955 012771, Fax:+34 955 013473
E-mail: guillermo.antinolo.sspa@juntadeandalucia.es